To

Jia-Ping Yan,
Director, Science Editor Development Department
World Journal of Clinical Pediatrics

Sir,

Ref.: Revision of Manuscript NO: 62468 World Journal of Clinical Pediatrics

**Correlation of Cardiac Troponin T levels with inotrope requirement, hypoxic-ischemic encephalopathy, and survival in asphyxiated neonates**

Many thanks for the review and constructive suggestions. We now revised the manuscript taking into consideration reviewers’ queries and Editor’s comments. The revised text is mentioned in red fonts for easy identification. A point by point reply to reviewer queries is also attached with this letter. Language editing is taken from Grammarly Software using American English. We hope the revision is acceptable. Please suggest any further clarifications are needed. Hoping to hear from you soon

Yours sincerely

Dr Ramesh Bhat Y

Corresponding author
Many thanks for the review and constructive suggestions

Reviewer’s Comments
Excellent paper on cardiac involvement in asphyxiated newborns. Would be interesting if you could analyze the values of CTnT during therapeutic hypothermia and its changes as a prognostic marker of final outcome

Thanks for encouraging words. We agree with your suggestion but unfortunately complete data on therapeutic hypothermia is not captured during this study. This issue we will consider in our next study. We write this as limitation in the present study.

Science editor 5 Issues
(1) The “Author Contributions” section is missing. Please provide the author contributions
Author contribution section is added now

(2) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor
For picture, PPT is now provided

(3) PMID and DOI numbers are missing in the reference list. Please provide the PubMed numbers and DOI citation numbers to the reference list and list all authors of the references. Please revise throughout PMID and DOI for references now provided and all authors are listed in the references

(4) The “Article Highlights” section is missing. Please add the “Article Highlights” section at the end of the main text
The article highlight section is now added at the end of the main text now

(5) The authors need to fill out the STROBE checklist with page numbers.
Strobe checklist is filled now with page numbers

Company editor-in-chief: I have reviewed the Peer-Review Report, full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Pediatrics, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors.
Many thanks

STEPS FOR SUBMITTING REVISED MANUSCRIPT
Step 1: Author Information
Please click and download the Format for authorship, institution, and corresponding author guidelines, and further check if the authors names and institutions meet the requirements of the journal.
Rechecked
Step 2: Manuscript Information
Please check if the manuscript information is correct.
Rechecked